Author’s response to reviews

Title: Toremifene, a selective estrogen receptor modulator, significantly improved biochemical recurrence in bone metastatic prostate cancer: a randomized controlled phase IIA trial

Authors:

Tetsuya Fujimura (tfujimura-jua@umin.ac.jp)
Satoru Takahashi (takahashi.satoru@nihon-u.ac.jp)
Haruki Kume (kume@kuc.biglobe.ne.jp)
Tomohiko Urano (turanotky@gmail.com)
Kenichi Takayama (knt56ta@hotmail.com)
Yuta Yamada (yyamada2029@gmail.com)
Motofumi Suzuki (uromotofumi@gmail.com)
Hiroshi Fukuhara (hifukuha@bk9.so-net.ne.jp)
Tohru Nakagawa (tohru-tky@umin.ac.jp)
Satoshi Inoue (sinoue07@gmail.com)
Yukio Homma (homma-uro@umin.ac.jp)

Version: 2 Date: 16 Oct 2015

Author’s response to reviews:

We appreciate the comments of the editor regarding our manuscript. Itemized responses to the editor’s comments are listed in the following.

Response

1. I have added all authors email address in the title page.
2. My draft was checked by native English speakers.
3. I have added that ST, SI, and YH revised the draft.
4. Our study was approved by ethical committee called internal review board at the University of Tokyo (approval number; P2008054) (In page 7 lines 4).